Author's response to reviews

Title: 16-year excess all-cause mortality of newly diagnosed type 2 diabetic patients: a cohort study

Authors:

Lars J Hansen (l.hansen@gpract.ku.dk)
Niels de Fine Olivarius (niels.olivarius@gpract.ku.dk)
Volkert Siersma (V.Siersma@gpract.ku.dk)

Version: 3 Date: 28 October 2009

Author's response to reviews:

Dear BioMed Central,

I have revised the manuscript according to your suggestions.

1) A Conclusion section has been inserted.

2) Reference 14 and 16 has been updated. Reference 16 = a report made and published by the Danish National Board of Health. Thus, I believe this is the correct way to cite this report.

3) Key word has been removed.

4) Figures has been improved.

5) All tracking has been removed form the manuscript.

Sincerely yours,

Lars J Hansen